

G OPEN ACCESS

Citation: de la Fuente J, Grotzinger AD, Marioni RE, Nivard MG, Tucker-Drob EM (2022) Integrated analysis of direct and proxy genome wide association studies highlights polygenicity of Alzheimer's disease outside of the APOE region. PLoS Genet 18(6): e1010208. https://doi.org/ 10.1371/journal.pgen.1010208

Editor: Michael P. Epstein, Emory University, UNITED STATES

Received: October 13, 2021

Accepted: April 19, 2022

Published: June 3, 2022

Copyright: © 2022 de la Fuente et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data that support the findings of this study are all publicly available or can be requested for access. Specific download links for various datasets are directly below. Summary statistics for IGAP are available from: https://www.niagads.org/datasets/ng00075 Summary statistics for UK Biobank can be downloaded from: https://www.ccace.ed.ac.uk/ node/335. **RESEARCH ARTICLE**

Integrated analysis of direct and proxy genome wide association studies highlights polygenicity of Alzheimer's disease outside of the APOE region

Javier de la Fuente^{1,2}*, Andrew D. Grotzinger^{1,3}, Riccardo E. Marioni⁴, Michel G. Nivard⁵, Elliot M. Tucker-Drob^{1,2}*

1 Department of Psychology, University of Texas at Austin, Texas, United States of America, 2 Population Research Center and Center on Aging and Population Sciences, University of Texas at Austin, Texas, United States of America, 3 Psychiatric and Neurodevelopmental Genetics Unit (PNGU) and the Center for Genomic Medicine, Massachusetts General Hospital, Boston, Massachusetts, United States of America, 4 Centre for Genomic and Experimental Medicine, Institute of Genetics and Cancer, University of Edinburgh, United Kingdom, 5 Department of Biological Psychology, VU University Amsterdam, the Netherlands

* j.delafuente@utexas.edu (JF); tuckerdrob@utexas.edu (EMT-D)

Abstract

Recent meta-analyses combining direct genome-wide association studies (GWAS) with those of family history (GWAX) have indicated very low SNP heritability of Alzheimer's disease (AD). These low estimates may call into question the prospects of continued progress in genetic discovery for AD within the spectrum of common variants. We highlight dramatic downward biases in previous methods, and we validate a novel method for the estimation of SNP heritability via integration of GWAS and GWAX summary data. We apply our method to investigate the genetic architecture of AD using GWAX from UK Biobank and direct casecontrol GWAS from the International Genomics of Alzheimer's Project (IGAP). We estimate the liability scale common variant SNP heritability of Clinical AD outside of APOE region at \sim 7–11%, and we project the corresponding estimate for AD pathology to be up to approximately 23%. We estimate that nearly 90% of common variant SNP heritability of Clinical AD exists outside the APOE region. Rare variants not tagged in standard GWAS may account for additional variance. Our results indicate that, while GWAX for AD in UK Biobank may result in greater attenuation of genetic effects beyond that conventionally assumed, it does not introduce appreciable contamination of signal by genetically distinct traits relative to direct case-control GWAS in IGAP. Genetic risk for AD represents a strong effect of APOE superimposed upon a highly polygenic background.

Author summary

In this article we demonstrate that common approaches for combining direct and proxy GWAS data produce dramatic underestimates of heritability, and we introduce a novel multivariate method for recovering unbiased estimates. Applying our method to direct

Funding: This work presented here would not have been possible without the enormous efforts put forth by the investigators and participants from the International Genomics of Alzheimer's Project (IGAP) and the UK Biobank. The work from these contributing groups was supported by numerous grants from governmental and charitable bodies. E.T.D. was supported by the National Institutes of Health (NIH; https://www.nih.gov/) grants R01AG054628 and RF1AG073593. J.F. and E.M.T-D. are members of the Population Research Center (PRC) and Center on Aging and Population Sciences (CAPS) at The University of Texas at Austin, which are supported by National Institutes of Health (NIH) grants P2CHD042849 and P30AG066614, respectively. R.E.M. was supported by Alzheimer's Research UK major project grant PG2017B-10 (https://www.alzheimersresearchuk. org/). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: R.E.M has received a speaker fee from Illumina and is an advisor to the Epigenetic Clock Development Foundation.

and proxy GWAS data for Alzheimer's disease we obtain estimates of liability-scale SNP heritability of AD outside of APOE region that are more than double the most recent estimates. Our analysis of local SNP heritability indicates that nearly 90% of common variant risk for Clinical AD represent a polygenic signal that is relatively diffusely distributed across the genome outside of the APOE region. We add substantial clarity to the scientific understanding the genetic architecture of AD, by introducing and validating new methodology that can be widely applied by researchers seeking to incorporate family information into GWAS.

Introduction

Genome-wide association studies (GWAS) of proxy-phenotypes using family history of disease (GWAX) can substantially boost power when combined with traditional case-control GWAS across a range of disease traits [1,2]. The benefits of GWAX for enhancing GWAS discovery have been particularly pronounced in the context of late-onset Alzheimer's Disease (AD) [3– 5], a neurocognitive disorder of aging that is clinically characterized by significant cognitive declines that interfere with independence in everyday activities, and biologically characterized by amyloid-predominant neuritic plaques, tau-predominant neurofibrillary tangles, and neurodegeneration [6-10]. Recent meta-analyses combining GWAX with direct GWAS have expanded the number of AD-relevant loci far beyond the well-established APOE variant, to 75 loci in total [5]. However, despite a twin-based heritability estimates of approximately 60% [11], these studies indicate very low common variant SNP heritability after excluding the APOE region, with the most recently reported estimate of 2.5% coming from the largest combined GWAX-GWAS meta-analysis of AD to date [4]. This estimate is noticeably lower than estimates obtained from the application of both LDSC and raw data-based methods in earlier studies that have only included direct case-control designs, albeit in smaller samples [12–14]. If valid, this low SNP heritability estimate may call into question the prospects of continued progress in genetic discovery for AD within the spectrum of common variants.

Here, we revisit the genome-wide architecture of AD. In scrutinizing recent approaches for estimating SNP heritability from GWAX, we find that even under scenarios in which the assumptions of the standard GWAX model are met, commonly used approaches for combining GWAX and GWAS data produce dramatic underestimates of SNP heritability. We develop a flexible data-driven multivariate framework for the accurate estimation of SNP heritability from GWAS and GWAX summary data, even in the presence violations of the standard GWAX assumptions. Using GWAS data from the International Genomics of Alzheimer's Project (IGAP) [12] and GWAX data from UK Biobank [3] our multivariate method yields substantially increased heritability estimates for clinically defined AD relative to the recently reported estimates from naïve meta-analysis of GWAS and GWAX data. We find that common genetic variants distributed outside of the APOE region account for between approximately 7% and 11% of variation in liability for Clinical AD. Informed by prevalence rates from epidemiological and molecular imaging data we project that common variants distributed outside of the APOE region may account for approximately 23% of variation in liability for biological AD (Alpha+ and Tau+). Our cumulative analysis of local SNP heritability of AD indicates a relatively continuous increase in genetic signal across the genome with a sharp discontinuity at the APOE locus, indicating that while the APOE region accounts for approximately 10% of common variant effects on AD, the remaining common variant risk for AD represents polygenic signal that is diffusely distributed across the genome.

Results

Overview of methods

Recent large scale meta-analyses combining direct-GWAS and GWAX data have used either inverse variance weighted meta-analysis of regression coefficients [3,15] or sample size weighted meta-analysis of Z statistics [4,16]. For the inverse variance approach, the standard correction factor for GWAX of the phenotype of a single first degree relative using the offspring genotype is to multiply regression coefficients and their SEs by 2.0 to correct for 50% attenuation due to 50% genetically relatedness (note that in the R^2 metric, this amounts to 75% attenuation). A similar correction to the Z statistics approach is possible (section 3 in S1 Supplementary Note) but not typically made. However, even with such corrections, estimation of SNP heritability from the summary statistics produced by either method produces estimates that are severely biased, unless the sample size input is further corrected (section 4 in S1 Supplementary Note). Moreover, GWAX estimates will be attenuated by more than the 50% assumed by the standard correction under a wide range of circumstances such as those in which: a proportion of genotyped individuals report on the phenotypes of their step or adoptive parents (such that average genetic relatedness of phenotyped and genotyped individuals falls below 50%); individuals are not well-informed about, misremember or confuse their parents' phenotype or disease status (such that heritability of the GWAX phenotype is attenuated, or contaminated by other heritable phenotypes); or the average diagnostic quality or criteria differ between proxy reports of historical disease status and direct GWAS of carefully screened case-control sample (such that heritability of the GWAX phenotype is attenuated). Here, we develop a multivariate model that directly estimates the appropriate correction empirically from the GWAX and direct GWAS summary data in order to produce unbiased estimates of SNP heritability and SNP effects without manual correction of effect size estimates, standard errors, or sample sizes.

Our multivariate model (section 5 in S1 Supplementary Note and S1 Fig) integrates summary data from three sources: direct GWAS, maternal GWAX, and paternal GWAX. In the model, the total genetic propensity toward AD risk is represented as a latent factor, *F*, that is specified to affect the direct GWAS phenotype and two GWAX phenotypes according to the following system of regression equations

Y _{direct}		λ_{direct}		u _{direct}	
Y_{mat}	=	λ_{mat}	F +	u _{mat}	,
Y _{pat}		λ_{pat}		u _{pat}	

where the λ coefficients relate *F* to measured phenotypes *y*, and the *u* terms are residual genetic propensities toward each of the measured phenotypes that are independent of *F*, and uncorrelated with one another and with *F*. We specify the model with the minimal identification constraint that $\lambda_{direct} = 1$ such that *F* takes on the scale of the direct GWAS phenotype, and σ_F^2 can be interpreted as an unbiased estimate of the SNP heritability of the meta-analyzed phenotype in the direct GWAS metric. Note that the standard GWAX approach[1] implicitly treats $\lambda_{mat} = \lambda_{pat}$ = .5 and $\sigma_{u_{Direct}}^2 = \sigma_{u_{Mat}}^2 = and \sigma_{u_{Pat}}^2 = 0$. Thus, under our more flexible multivariate parameterization in which these terms are freely estimated, the departure of λ_{mat} and λ_{pat} from .5 and departure of the σ_u^2 terms from 0 indicate departure of the empirical data from the standard GWAX assumptions. If we are willing to assume that the direct GWAS represents the pure signal of interest, uncontaminated from other heritable traits (e.g. other forms of dimension), we can fix $\sigma_{u_{Direct}}^2$ to 0, in which case the multivariate model becomes similar to the MTAG model [17,18]. To estimate meta-analytic summary statistics using this multivariate model (section 6 in <u>S1</u> <u>Supplementary Note</u>), we expand it to include the effect of an individual genetic variant, x, on the F as follows

$$F=\gamma x+e,$$

where γ is an unstandardized regression coefficient and *e* is a residual. As described in section 7 of in <u>S1 Supplementary Note</u>, a SNP-specific heterogeneity statistic (Q_{SNP}) is computed to gauge the extent to which the regression coefficient γ does not well account for the pattern of associations between that SNP and the individual GWAS and GWAX.

Models are estimated in Genomic SEM [18] using a two-stage approach. In the first stage, the empirical liability-scale genetic covariance matrix and its sampling covariance matrix are estimated. In the second stage, the model is fit to the matrices using the diagonally weighted least squares (WLS) fit function with sandwich correction, as described in Grotzinger et al. [18]. Beyond its flexible capabilities for user-specified modeling, Genomic SEM is advantageous for its accommodation of unknown and varying degrees of sample overlap. This may be particularly advantageous for instances in which individuals included in the GWAX are related to those in the direct GWAS, or when users seek to incorporate GWAX from relatives who may themselves be related (e.g. both brothers and sisters of the directly genotyped individuals [18]). We refer to our multivariate model as a Genomic SEM Relaxed Model, in that it uses Genomic SEM to relax the conventional assumption that $\lambda_{mat} = \lambda_{pat} = .5$. A complete tutorial for our method, including detailed code, can be found in the Tutorials tab of https://github.com/GenomicSEM/GenomicSEM/wiki.

Simulation results

We simulated genome-wide summary statistics for direct GWAS and maternal and paternal GWAX manipulating the attenuation coefficients (λ) for maternal and paternal GWAX assuming either continuous or binary traits. In experimental condition 1, the simple GWAX model assumptions hold (i.e., $\lambda_{mat} = \lambda_{pat} = .5$). In experimental condition 2, we consider a strong deviation from the simple GWAX model, (i.e., $\lambda_{mat} = \lambda_{pat} = .25$). In experimental condition 3 we set the attenuation coefficients to mimic those empirically estimated from our multivariate model using real GWAS and GWAX data (i.e., $\lambda_{mat} = .463$, $\lambda_{pat} = .366$). Although our simulations refer to maternal and paternal GWAX, it is of note that our multivariate model can incorporate family history GWAX data across diverse degrees of genetic relatedness. We compared the Genomic SEM approach to a conventional uncorrected approach in which GWAS and GWAX summary data are meta-analyzed without explicit consideration of whether they were derived from direct or proxy data sources (cf. by Jansen et al. [16] and Wightman et al. [4]) and the conventional corrected approach (reflective of those taken by Marioni et al. [3], and Bellenguez et al. [5].

The top row of Figs 1 and 2 presents simulation results for continuous phenotypes (Fig 1) and binary phenotypes (Fig 2) with respect to heritability estimates from conditions in which sample size is held constant and the λ coefficients vary, reflecting different degrees of deviation from the simple GWAX assumption that $\lambda_{mat} = \lambda_{pat} = .5$. Note that liability scale heritability estimates are reported for binary phenotypes. Supplemental results for the recovery of SNP heritability for continuous traits in conditions with varying sample size are reported in S11 Table and displayed in S2 Fig. The multivariate Genomic SEM approach provided essentially unbiased heritability estimates (h_F^2) across all conditions, outperforming the conventional (uncorrected) approach in all conditions, and outperforming the conventional approach with standard correction, and the conventional approach



Fig 1. Simulation Results for continuous phenotypes. Distribution of SNP heritability estimates (top row) and individual SNP effects (bottom row) for Conventional (Uncorrected), Conventional (Standard Correction), and Genomic SEM Relaxed Model across conditions. The vertical dashed red lines indicate the true parameter value in the population. Complete simulation results for all conditions for continuous traits are reported in S2, S3, S10 and S11 Tables.

https://doi.org/10.1371/journal.pgen.1010208.g001

with standard and liability correction in conditions in which the assumptions of the simple GWAX model did not hold, for both continuous (Percent Bias Error, %BE, range for conventional [uncorrected] approach: 76.97%-88.91%; %BE range for conventional [standard correction]: 21.39%-55.56%; %BE range Genomic SEM relaxed model: 0.04%-0.36%), and binary traits (%BE, range for conventional [uncorrected] approach: 75.93%-86.94%; %BE range for conventional [standard correction]: 61.68%-71.85%; %BE range for conventional [standard + liability correction]: 10.82%-34.48%; %BE range Genomic SEM relaxed model: 1.69%-2.65%). As expected, the conventional approach with the standard correction (for continuous traits), and the conventional approach with standard and liability corrections (for binary traits), performed similarly to the Genomic SEM approach when the standard assumption that $\lambda_{mat} = \lambda_{pat} = .5$ held. Bias in the conventional approaches was related to the extent to which the population values of λ_{mat} and λ_{pat} deviated from .5.

The bottom row of Figs 1 (continuous traits) and 2 (binary traits) presents simulation results with respect to individual SNP effects for conditions in which sample size is held constant and the λ coefficients vary. Supplemental results for recovery of individual SNP effects for continuous traits with varying sample size across conditions are provided in S12 Table and S3 Fig. Results for the recovery of individual SNP effects were consistent with those for the recovery of heritability. Our multivariate Genomic SEM approach exhibited consistently unbiased performance across all conditions (%BE range 0.32%-1.49%), outperforming the conventional uncorrected and corrected approaches in all conditions for continuous traits (Percent Bias Error, %BE, range for conventional [uncorrected] approach: 64.19%-78.33%; %BE range for conventional [standard correction]: 1.01%-33.46%; %BE range Genomic SEM relaxed model: 0.23%-0.87%), and the conventional uncorrected and corrected approach in conditions



Fig 2. Simulation Results for binary phenotypes. Distribution of liability-scale SNP heritability estimates (top row) and individual SNP effects (bottom row) for Conventional (Uncorrected), Conventional (Standard Correction), Conventional (Standard + Liability Correction), and Genomic SEM Relaxed Model across conditions. The vertical dashed red lines indicate the true parameter value in the population. Complete simulation results for all conditions are reported in S2 and S3 Tables.

https://doi.org/10.1371/journal.pgen.1010208.g002

in which the assumptions of the simple GWAX model did not hold for binary traits (Percent Bias Error, %BE, range for conventional [uncorrected] approach: 50.40%-63.44%; %BE range for conventional [standard correction]: 5.88%-18.75%; %BE range Genomic SEM relaxed model: 0.29%-0.59%). When the simple GWAX model assumptions held, the Genomic SEM approach and the conventional approach with the standard correction were both unbiased, but the conventional approach without correction remained biased. Bias in the conventional approaches was related to the extent to which the population values of λ_{mat} and λ_{pat} deviated from .5.

We additionally simulated GWAS and GWAX summary statistics under a null scenario in which genetic covariances among GWAS and GWAX were set to 0, which we analyzed using the multivariate Genomic SEM model. In this scenario, the GWAS and GWAX represent genetically distinct phenotypes, such that a factor representing their shared genetic architecture is expected to produce a heritability estimate of 0. As expected, the mean SNP heritability across replications was 0%, indicating an unbiased parameter estimate. Moreover, SNP heritability was significant at p < .05 in none of hundred replications (S9 Table), compared to an expected frequency of 5, indicating appropriate, if not conservative, Type-I error control.

Multivariate model of direct GWAS and GWAX of Alzheimer's disease

We applied our multivariate model to empirical summary data from the direct case-control GWAS of AD in IGAP [12] and GWAX of maternal and paternal AD in UK Biobank [3]. Key descriptive statistics for these three contributing datasets are reported in the top portion of Table 1, and multi-trait LDSC results (cross trait intercepts and genetic correlations) are

					DSC ^a									
	Phenotype	Mean χ^2	λ_{GC}	h ² liability (prevalence = 5%)	SE	r	Intercept	Maternal Cases	Maternal Controls	Paternal Cases	Paternal Controls	Direct Cases	Direct Controls	N reported
Contributing	IGAP—Stage I	1.117	1.089	0.073	0.012	6.083	1.025	1				21,982	41,944	63,926
Summary Data	UKB Maternal	1.049	1.032	0.017	0.004	4.250	1.006	27,696	260,980			,		288,676
	UKB Paternal	1.031	1.029	0.012	0.006	2.000	1.009	,		14,338	245,941	,	1	260,279
Meta-analyses	Multivariate Model	1.139	1.079	0.069	0.008	8.625	1.001	27,696	260,980	14,338	245,941	21,982	41,944	ı
	Marioni et al. (2019)	1.142	1.085	0.023	0.003	7.333	1.025	27,696	260,980	14,338	245,941	25,580	48,466	388,324
	Jansen et al. (2019)	1.114	1.068	0.020	0.002	9.500	1.000	47,793 ^b	328,320 ^b	ı	1	24,087	55,058	455,258
^a LDSC-derived res	ults are restricted t	to HapMá	ıp3 SNP	's only, excluding the M	HC and	l APOE	(chr:19, bp	45,116,911: 46	,318,605) regio	ns. LDSC-der	ived results ar	e derived fr	om for the m	ultivariate
brhese Ns refer to t	1 by submitting the the total numbers of	e summaı of proxy (ry statist cases and	ics from our multivaria d proxy controls (i.e., m	te meta laternal	-analysi + pater	s to LDSC. nal cases and	d maternal + p	aternal contro	ls).				
<i>Note</i> : Heritability e	stimates reported	here are (on a liab	ility scale, based on a po	opulatic	n preva	lence of 5%.	Elsewhere in	this paper, we	consider sensi	tivity of liabili	ty scale heri	itability estim	ates to

Table 1. Descriptive statistics for contributing cohorts and previous meta-analyses of case-control and proxy-phenotypes of Alzheimer's disease and LDSC output.

other values, and to the possibility of undiagnosed AD within control participants, and we specifically compare estimates to those obtained with a population prevalence of 5% used by Wightman et al. [4] to produce their estimate of h² liability of 2.5%. Heritability presented for GWAX phenotypes and Marioni and Jansen meta-analyses are naïve, and have not been corrected for the indirect nature of the GWAX.

https://doi.org/10.1371/journal.pgen.1010208.t001

Standardized

provided in S4 Fig. Note that all LDSC estimates reported in Table 1 and S4 Fig are derived from LD scores based on common variants (MAF \geq .05) outside of the MHC and *APOE* regions, using the AD population prevalence rate of 5%. Cross-trait intercepts among IGAP and the maternal and paternal GWAX were all ~0, indicating that there was effectively no sample overlap or dependency between LDSC estimates. Genetic correlations were close to 1.0 between IGAP and maternal AD (rG = 0.915, SE = 0.153, *p* < 0.001) and paternal AD (rG = 0.842, SE = 0.206, *p* < 0.001) in UKB. The genetic correlation between maternal and paternal AD was also very high (rG = 0.815, SE = 0.253, *p* = 0.001).

In the bottom portion of Table 1 we provide key descriptive of summary statistics from our multivariate meta-analysis and from two recent meta-analyses that have combined GWAS and GWAX AD data [3,16]. Estimates from the full multivariate model of AD are displayed in Fig 3. In the model itself, the liability-scale SNP heritability estimate was 6.95% (SE = 3%) assuming a population prevalence of 5%, over double that estimated by application of LDSC to the summary statistics from the other two recent AD meta-analyses, using their reported sample sizes, assuming the same population prevalence. A version of the model that constrained residual variance of the direct GWAS to 0 (reflecting the assumption that the direct GWAS does not contain ancillary genetic signal unrelated to AD) did not produce a significant decrement in model fit ($\chi^2(1) = .021$, p = .89) and produced a slightly larger liability scale SNP heritability estimate (7.17%) with a substantially smaller standard error (SE = 1.21%).

Applying LDSC to GWAS summary statistics produced by our multivariate method produced a very similar liability-scale SNP heritability estimate (6.9%, SE = .8%; Table 1; assuming a 5% population prevalence rate) to that produced by the model itself. In the full model, the unstandardized loading for maternal AD (λ_{mat} = .463) was slightly attenuated with respect to



Unstandardized

Fig 3. Unstandardized (left) and standardized (right) empirical results from multivariate genetic analysis of Alzheimer's disease. The liabilityscale SNP heritability estimate of 6.9% is on the scale of the direct GWAS, and is for common variants (MAF \geq .01) outside of the MHC and *APOE* regions, using the AD population prevalence of 5%. *u* = residual genetic variance.

https://doi.org/10.1371/journal.pgen.1010208.g003

the expectation under the standard GWAX model ($\lambda_{mat} = .5$). The unstandardized loading for paternal AD ($\lambda_{pat} = .366$) reflected approximately 27% attenuation of regression effects (i.e.[.5-.366]/.5) and approximately 46.42% attenuation of R² and liability-scale h² estimates ([.5²-.366²]/[.5²]) relative to the expectation under the standard model ($\lambda_{pat} = .5$). Allowing the λ_{mat} and λ_{pat} parameters to be freely estimated can avoid the potential for bias stemming from violations of the standard assumption, in this case particularly for paternal effects, when estimating genome wide meta-analytic summary statistics. Residual variances for the direct GWAS and both GWAX were trivial, indicating that the GWAS and GWAX were not tapping genetically distinct phenotypes to an appreciable extent. We note that the liability-scale heritability estimates reported in this section are based on an assumed AD population prevalence rate of 5%, as per the LDSC analyses by Wightman et al. [4]. Next, we consider sensitivity of the heritability estimate to different assumptions regarding the sample and population prevalence rates of AD.

Liability scale heritability across a range of assumed population prevalence rates

Heritability estimates for case-control traits, such as AD, are based on a liability threshold model, which assumes a continuously distributed liability toward the binary phenotype in the population. Estimates of liability-scale SNP heritability are sensitive to assumptions about the lifetime prevalence of the disorder in the population and the extent to which unaffected individuals have been successfully screened for the disorder (see Methods). When population prevalence rates are high or when control participants are unscreened, differences in allele frequencies between cases and controls represent less extreme comparisons along regions of the liability distribution. In such circumstances, the inferred liability scale heritability is higher than would be inferred from the same case-control difference in allele frequencies for a disorder with a lower population prevalence or when control participants have been carefully screened. This issue is particularly germane to the study of AD, a disorder whose: (i) clinical prevalence rate increases from less than 1% in middle adulthood to approximately 30% by old age [19], (ii) that is known to go undetected at high rates for decades prior to diagnosis due to ancillary factors (e.g. educational attainment) unrelated to biological severity [20], and (iii) whose pathophysiological basis may be more than twice as prevalent as its clinical diagnosis at any given age [10].

To gauge the effects of assumptions regarding population prevalence on our liability-scale SNP heritability estimate for AD, we varied the assumed population prevalence rate. We provide rough age equivalents of these prevalence rates based on published epidemiological data for Clinical AD [19], assuming that control participants were appropriately screened. We refer to this estimate as the estimate of heritability of Clinical AD, in that this estimate does not account for biological AD that is likely to exist in a potentially sizable subset of control participants. Estimates from the full multivariate model are displayed in the left panel of Fig 4 (estimates from the application of LDSC to the summary statistics from our multivariate metaanalysis were very similar; S7 Fig). Heritability estimates of Clinical AD increase as the population prevalence rate increases, reaching approximately a liability-scale $h^2 = 12\%$ at prevalence = 30%. We note that Wightman et al. [4], estimate a liability-scale heritability of 2.5% assuming a population prevalence rate of 5%. At this same prevalence rate, the estimate from our full multivariate model is over double (liability $h^2 = 6.95\%$) that of Wightman et al. This is likely a reflection of the fact that Wightman et al. do not appear to make any correction to their LDSC estimate to account for any attenuation in the GWAX estimates due to their indirect nature.



Fig 4. Estimated common variant liability-scale SNP heritability of AD (outside of the MHC and *APOE* regions) according to different assumptions regarding the population prevalence of Clinical AD and Biological AD (Alpha+ and Tau+). We provide rough age equivalences for each prevalence rate on the top x axis. The purple diamond represents the estimate of 2.5% by Wightman et al. [4], which was based on an assumed population prevalence rate of 5%. The yellow diamond represents the estimate from the multivariate model introduced here, using the same assumed population prevalence rate of 5% (Clinical AD h² liability = 0.069; Biological AD h² liability = 0.073). The shaded area around the line reflects +/- 1 SE of the h² estimate. The steeper shift in the SNP heritability of AD for biological AD compared to clinical AD as a function of population prevalence rate of biological AD increases, the extent of case contamination in control participants increases, and the correction for undetected AD in control participants produces more dramatic increases in the projected heritability. Biological AD prevalence rates are from recently published positron emission tomography data [10].

https://doi.org/10.1371/journal.pgen.1010208.g004

We also estimated liability-scale SNP heritability (for common variants outside of the MHC and APOE regions) across a range of different assumed prevalence rates of biological AD, employing a correction for case contamination (see Methods) in control participants (who were primarily screened for Clinical AD but not AD pathology) and provide rough age equivalents of these prevalence rates for biological AD (Alpha+ and Tau+) from recently published positron emission tomography data [10]. We refer to this estimate as the projected estimate of biological AD heritability. Results are displayed in the right panel of Fig 4. Heritability estimates increase steeply as the prevalence rate increases, reaching approximately a liability-scale $h^2 = 23\%$ at a population prevalence of 30%. The steeper shift in the SNP heritability of AD for biological AD compared to clinical AD stems from our correction for undetected biological AD within control participants who have primarily only been screened for clinical AD. As the assumed prevalence rate of biological AD increases, the extent of case contamination in control participants increases, and the correction for undetected AD in control participants produces more dramatic increases in the projected heritability. The validity of this inference relies on the assumptions that individuals with clinical AD diagnoses are representative of the larger set of individuals with biological AD, and do not constitute a subgroup of those with more severe biological AD. This assumption is supported by a large body of work indicating that ancillary factors, such as educational attainment, are associated with clinical AD rates among individuals with equivalent levels of brain pathology [20].

Liability scale heritability using stratified LDSC

To allow for potentially uneven contributions of SNPs to heritability across biologically, evolutionarily, and MAF defined categories, we fit our Genomic SEM relaxed model to a genetic covariance matrix derived using stratified LDSC [21] based on 97 baseline annotations [22], assuming 5% population prevalence. We obtained larger, albeit less precise, estimates of liability-scale SNP heritability ($h^2 = 8.94\%$, SE = 5.17%) than those obtained when using the standard LDSC model ($h^2 = 6.95\%$, SE = 3%).

Inferring common variant polygenicity from local SNP heritability

Our results indicated relatively high common variant SNP heritability of AD outside of the *APOE* locus. To investigate whether this SNP heritability was attributable to a small number of genomic regions, or distributed more evenly across the genome, we used Heritability Estimation from Summary Statistics (HESS) [23] to estimate local common variant SNP heritability in 1,703 approximately independent blocks across the genome using the summary statistics from our multivariate meta-analysis. Following Sinnott-Armstrong et al. [24], we plot the proportion cumulative heritability of AD liability across the genome in Fig 5. Because HESS allows for negative estimates of local SNP heritability, estimation error is not expected to produce increases in cumulative heritability; only true polygenic signal will produce such increases. It can be seen that heritability accumulates relatively continuously across the entirety of the genome, with a pronounced discontinuity at the *APOE* locus on chromosome 19. This locus accounts for 10.53% of total common variant SNP heritability is attributable to genetic signal independent of *APOE*. Local SNP heritability in the other loci that were genome-wide significant in



Fig 5. Proportion of cumulative heritability of Clinical AD across the genome, as estimated with HESS (24). Yellow triangles represent genome-wide significant loci from the multivariate GWAS of AD.

https://doi.org/10.1371/journal.pgen.1010208.g005

the multivariate GWAS was strongly correlated with their GWAS effect sizes (S8 Fig) and accounted for an additional 9.46% of the total common variant liability-scale SNP heritability, leaving 79.98% of the total common variant liability-scale SNP heritability unexplained by genome-wide significant loci. The observations that heritability accumulates relatively continuously outside of the *APOE* locus and that substantial proportion of SNP heritability remains outside of genome-wide significant loci suggests that genetic risk for AD may be affected by core pathways superimposed on a more diffuse polygenic background. Using the conservative estimate of population prevalence of 5% the total heritability of AD on the liability scale as estimated with HESS was 11.09% (9.92% excluding the *APOE* locus). Similar patterns were observed when HESS analyses were limited to GWAS data from IGAP only (S9 Fig).

Multivariate results for individual SNP effects

We estimated meta-analytic summary statistics containing individual SNP effects on AD in our multivariate model using Genomic SEM, a Manhattan plot for which is provided in S5 Fig, and qqplots for which are provided in S6 Fig. The mean χ^2 for the common factor GWAS output was 1.139. The mean $\chi^2(1)$ for Q_{SNP} was 1.008, and there were no genome-wide significant ($p < 5 \times 10^{-8}$) hits for Q_{SNP}, indicating little evidence for genome-wide heterogeneity in SNP effects across direct GWAS, maternal GWAX, and paternal GWAX after the empirically derived attenuation coefficients (λ) are taken into account. The LDSC intercepts were all very close to 1.0, indicating that inflation of test statistics was predominately attributable to true polygenic signal rather than population stratification (Table 1).

We identified 282 independent significant SNPs and 93 lead SNPs in a total of 24 genomewide significant loci associated with AD (S4–S8 Tables). All of these significant loci were previously reported in published meta-analysis of GWAX and direct GWAS of AD by Marioni et al. [3], Jansen et al. [16], or Schwartzentruber et al. [25]. For the genome-wide significant loci, we computed meta-analytic estimates using the inverse variance weighted approach and the Z approach. We applied each approach both naively (i.e. without correction) and with a correction for attenuation due to the indirect nature of the GWAX with the standard correction (S6 Table). As expected, based on the fact that the empirically derived λ coefficients from our model were close to .5, we found that that the inverse variance weighted approach with the standard correction produced effect size estimates similar to those from our meta-analytic model. In contrast, the uncorrected approaches produced substantially deflated effect size estimates. The Z Statistic approach, even with the standard correction, still tended to produce somewhat deflated effect size estimates. This can be attributed to the fact that the correction employed corrected for attenuation due to the indirect nature for the GWAX, but did not correct for variability in prevalence rates stemming from the ascertained nature of the samples.

Discussion

Recent reports of very low SNP heritability of AD have called into question the prospects of continued progress in genetic discovery for AD within the spectrum of common variants. We have demonstrated that common methods used to produce these estimates are dramatically downwardly biased. We have introduced and validated a novel multivariate method for the joint analysis of direct GWAS and proxy GWAX summary data that relaxes standard assumptions and recovers unbiased estimates of common variant SNP heritability and of individual SNP effects under a variety of conditions. Compared to other naïve methods that boost power for estimating individual SNP effects on one phenotype by incorporating GWAS data from correlated phenotypes, our multivariate model specified within Genomic SEM is a formal model of multivariate genetic architecture from which both interpretable genome-wide

parameters of interest, and individual SNP-specific effects can be estimated. Additionally, our model is unique in providing indices for violation of assumptions (e.g. Q_{SNP}, see section 7 in <u>S1 Supplementary Note</u>), which may safeguard against cross-contamination of genome-wide signals across correlated traits.

Using recently released GWAS and GWAX summary data for AD, we find some modest deviations of the patterning of genetic sharing from that expected under the standard GWAX model. In particular, whereas the standard GWAX model assumes that genetic effects are attenuated by 50% (in regression units; i.e. by 75% in R^2 units) relative to direct GWAS, we estimate somewhat greater attenuation: 54% (i.e. 1-.463) for maternal GWAX and 63% (i.e. 1-.366) for paternal GWAX. Interestingly, residual genetic variance specific to the individual GWAX was trivial, and there was no evidence of heterogeneity (Q_{SNP}) of individual SNP effects beyond that expected based on our multivariate model. Taken together, these results suggest that GWAX for Clinical AD in UK Biobank may result in greater attenuation of genetic effects beyond that conventionally assumed, but does not introduce appreciable contamination of signal by genetically distinct traits relative to direct GWAS in IGAP.

Our multivariate method estimates the common variant liability-scale SNP heritability of AD excluding the *APOE* region to be more than double the recent estimate of 2.5% by Wightman et al. [4], when using same assumptions regarding population prevalence. This estimate rises to as high as 11% for Clinical AD and 23% for biological AD when considering prevalence rates indicated by collateral epidemiological and neuroimaging data. Of course, the estimated range of plausible values for the heritability of biological AD are based on assumptions regarding the extent to which the genetic signal represented in individuals diagnosed with clinical AD can be extrapolated to individuals with latent biological AD. Future research into the genetic architecture of biological AD will benefit from the large scale genomic analysis of direct measures of AD pathophysiology using, for example, positron emission tomography [10].

Analysis of local SNP heritability indicates that almost 90% of common variant SNP heritability of AD exists outside of the *APOE* region. We find that this remaining heritability accumulates relatively continuously across the remainder of the autosome, indicating that a portion of genetic risk for AD is characterized by a relatively diffuse polygenic architecture. This pattern resembles that recently observed for three molecular traits whose biological pathways are well-understood [24]. These authors inferred that core gene sets representing proximal biological mechanisms play a sizable role in each trait, but that "most of the SNP-based heritability is driven by a massively polygenic background." Our results suggest that the same may pertain to the genetic architecture of AD, as has been generally postulated for complex traits by the omnigenic model [26].

Background polygenic architecture may reflect biological mechanisms underlying the pathophysiology of AD or ancillary factors that affect the likelihood of diagnosis of Clinical AD without contributing to disease onset or progression. The strongest candidates for such ancillary factors may be educational attainment and premorbid cognitive functioning, both of which longitudinal research indicates are not related to the onset or rate of cognitive decline but are related to the likelihood that declines will be detected by existing diagnostic protocols [7,27]. Genetic correlation analyses indicated that while these factors are indeed genetically correlated with AD, less than approximately 4% of SNP heritability of AD is explained by either factor, indicating that they are not by themselves sufficient to account for the polygenic component of AD genetic architecture. Longitudinal genomic research of neurocognitive change preceding and predicting eventual clinical AD will be of particular value for identifying pathways that may contribute to the polygenic risk for AD pathophysiology.

It is important to consider that the estimates of common variant SNP heritability from our multivariate modeling are based on the integration of summary data across many different

cohorts. Between-cohort variability in the genetic architecture of AD across cohorts, and in methods for ascertainment and diagnosis can lead to attenuation of heritability during the aggregation process. Indeed, raw data based estimates of heritability using more homogeneous, yet smaller, cohorts [28] have often produced even larger heritability estimates than those reported here. This discrepancy could also be explained by the distribution of SNP effects of AD. If the core pathways for AD consist of a small number of causal variants with relatively large effects, then approaches that rely on assumptions of high polygenicity and normally distributed SNP effects may be yield underestimates of SNP heritability.

Our primary models relied on genetic covariance estimates derived using the standard LDSC model, which assumes homogeneous contributions to trait heritability across SNPs. Importantly, liability scale SNP heritability estimates continued to be sizable when alternative estimation methods were used. For instance, HESS allows for heritability to vary across different regions of the genome. The liability scale SNP heritability estimate obtained by applying HESS to the summary statistics produced by our multivariate model was 11.09% (9.92% excluding the APOE locus), assuming a population prevalence of 5%. Moreover, stratified LDSC allows for potentially uneven contributions of SNPs across biologically, evolutionarily, and MAF defined categories. When we fit our multivariate model to a genetic covariance matrix derived using stratified LDSC using 97 annotations, we obtained an estimate of liability scale SNP heritability of 8.94%, assuming a population prevalence of 5%. Importantly, we followed the standard practice of estimating genetic covariances using common variants. Rare variants not tagged in standard GWAS are likely to account for additional variance beyond the heritability estimates reported here. Whether the genetic covariance structure across GWAS and GWAX phenotypes differs for rare variants compared to common variants is an open question.

Methods

Bias in conventional meta analyses combining GWAS and GWAX

Recent large scale meta-analyses combining direct-GWAS and GWAX data have used either inverse variance weighted meta-analysis of regression coefficients [3,29] or sample size weighted meta-analysis of Z statistics [4,16,30]. For the inverse variance weighted approach, it is well-known that because of the indirect nature of the GWAX, the regression coefficients and associated standard errors (the squares of which represent the sampling variances used for inverse variance weighting) must be multiplied by a correction factor prior to meta-analysis [1]. The standard correction factor for GWAX of the phenotype of a single first degree relative using the offspring genotype is 2 to correct attenuation due to 50% genetically relatedness. Although not commonly implemented in the literature, we show in section 3 in S1 Supplementary Note that a correction must also be made when implementing the Z statistic approach. We show that for a continuous trait, the sample size must be divided by the square of the correction factor (e.g. by 4 for the standard correction) when implementing the Z statistic approach, with further corrections to sample size needed for case-control traits in ascertained samples. Naïve analysis of GWAX summary statistics produces SNP heritability estimates that are downwardly biased by 75% or more. This bias will carry forward in meta-analysis combining direct GWAS with GWAX statistics (see section 4 in S1 Supplementary Note and simulation below). This will occur when estimating SNP heritability from summary statistics even when regression coefficients and SEs have been corrected, because methods for estimating SNP heritability typically rely on the ratio of the regression coefficients to their standard errors (i.e. the Z statistics), which is preserved under the correction. We show that entering the effective sample size (obtained by dividing the observed N by the square of the correction factor)

will produce the unbiased SNP heritability estimate when the appropriate correction factor is known (if the phenotype is a case-control trait, a conversion of the heritability estimate from the observed scale to the liability scale is still called for).

Whether the standard correction factor of 2.0 for GWAX regression coefficients is appropriate, and if not, how the appropriate correction factor can be identified is an unaddressed topic. The standard correction factor is derived simply on the basis of the 50% attenuation to regression estimates expected when proxy cases are first degree relatives of the genotyped individuals. However, attenuation will be more severe, such that the standard correction is insufficient, under a wide range of circumstances, such as those in which: a proportion of genotyped individuals report on the phenotypes of their step or adoptive parents (such that average genetic relatedness of phenotyped and genotyped individuals falls below 50%), when individuals are not well-informed about, misremember or confuse their parents' phenotype or disease status (such that heritability of the GWAX phenotype is attenuated, or contaminated by other heritable phenotypes), or when the average quality of the diagnostics is of lower quality for parental history reports than for direct GWAS of carefully screened case-control sample (such that heritability of the GWAX phenotype is attenuated, or contaminated by other heritable phenotypes). We derive the expected attenuation bias to both regression coefficients and heritability and genetic covariance estimates analytically in section 3 and 4 in <u>S1 Supplementary Note</u>. Below, we introduce a data-driven multivariate approach for meta-analyzing GWAX and direct GWAS summary data that estimates the amount of attenuation in the GWAX (and thus the appropriate correction) directly from the data. Our approach formally models the genetic covariance structure of the GWAX and direct GWAS summary data in order to produce unbiased estimates of SNP heritability and individual SNP effects without manual correction of effect size estimates, standard errors, or sample sizes.

A relaxed multivariate model for combining GWAS with GWAX

We use Genomic SEM to estimate a multivariate model of genetic risk for AD using summary data from three sources: direct GWAS, maternal GWAX, and paternal GWAX. In our multivariate model, the total genetic propensity toward AD risk is represented as a latent factor, *F*, that is specified to affect the direct GWAS phenotype and two GWAX phenotypes according to the following system of regression equations

$$egin{aligned} Y_{direct} \ Y_{mat} \ Y_{pat} \end{bmatrix} &= egin{bmatrix} \lambda_{direct} \ \lambda_{mat} \ \lambda_{pat} \end{bmatrix} F + egin{bmatrix} u_{direct} \ u_{mat} \ u_{pat} \end{bmatrix}, \end{aligned}$$

or more compactly as

$$Y = \Lambda F + U$$

where Λ is a vector of coefficients relating *F* to measured phenotypes *Y*, and U constitutes residual genetic propensities toward each of the measured phenotypes that are independent of *F*, and uncorrelated with one another and with *F*. This model is represented as a path diagram in S1 Fig (excluding the dashed portion).

This model implies that the genetic covariance matrix for Y_{Direct} , Y_{Mat} , and Y_{Pat} is given as

$$\Sigma = \Lambda \Psi \Lambda' + \Theta$$

where Ψ represents the covariances among the factors (and in this case contains a single element, representing the variance of F, i.e. σ_F^2), and Θ represents the covariances among the residuals, U (in this case a 3×3 diagonal matrix with diagonal elements $\sigma_{u_{Direct}}^2$, $\sigma_{u_{Mat}}^2$, and $\sigma_{u_{Par}}^2$).

We specify the model with the minimal identification constraint that $\lambda_{direct} = 1$ such that F takes on the scale of the direct GWAS phenotype, and σ_F^2 can be interpreted as an unbiased estimate of h_{SNP}^2 of the meta-analyzed phenotype in the direct GWAS metric. Under this parameterization the departure of λ_{mat} and λ_{pat} from .5 indicates departure of the empirical data from the standard GWAX model. We also consider an alternative parameterization in which we specify the model with the minimal identification constraint that $\sigma_F^2 = 1$, such that the variance of the latent factor F is standardized, and the freely estimated term λ_{Direct} can be interpreted as an unbiased estimate of $\sqrt{h_{SNP}^2}$ of the meta-analyzed phenotype in the direct GWAS metric. Under this parameterization, departure of $\lambda_{mat}/\lambda_{direct}$ and $\lambda_{pat}/\lambda_{direct}$ from .5 indicates departure of the empirical data from the standard GWAX model. Both parameterizations produce equivalent, just identified models, with 0 degrees of freedom, but differ in how parameters must be interpreted. These models can straightforwardly be extended to estimate genetic correlations between *F* and external GWAS phenotypes, such as educational attainment.

Estimation of SNP effects for multivariate GWAS

The multivariate model can be expanded to include the effect of an individual genetic variant, x, on the latent factor, F (S1 Fig, including the dashed portion). Such a model comprises the following two sets of equations:

$$Y = \Lambda F + U,$$
$$F = \gamma x + e,$$

where γ is an unstandardized regression coefficient and *e* is a residual. Such a model can be run for all available SNPs, one at a time, such that a complete set of meta-analytic summary statistics for F (in this case, AD), can be produced from the relaxed multivariate model. We use the minimal identification constraint $\lambda_{direct} = 1$, such that γ takes on the scale of the direct GWAS. To avoid the potential for variability in the optimization of the Λ and Θ parameters that may obscure interpretation of the γ coefficients across SNPs, we fixed these parameters to their values from the model without SNPs when estimating the model for each SNP in our empirical analysis. The resulting summary statistics from this multivariate GWAS serve as an alternative to those produced by more constrained Z statistic-based and (both corrected and uncorrected) inverse variance- based approaches. When contributing GWAS and GWAX are independent, as is the case for the empirical application presented here, the Effective N associated with the summary statistics from our multivariate model is calculated as Neff_{Direct} + Neff- $P_{aternal}^* \lambda_{Paternal}^2 + Neff_{Maternal}^* \lambda_{Maternal}^2$, where the λs are unstandardized factor loadings from the model. For summary statistics derived from a GWAS or GWAX meta-analysis, Neff = $\sum 4v_k(1-v_k)n_k$, and v_k is the sample prevalence for contributing study k. See Grotzinger et al. [31] for further detail.

Model estimation

Models are estimated in Genomic SEM [18] using a two-stage approach. In the first stage, the empirical liability-scale genetic covariance matrix S and its sampling covariance matrix V are estimated using a multivariate version of LDSC. When the model to be fit includes individual SNP effects, the S matrix is expanded to include a vector of genetic covariances between the SNP and each of the GWAS and GWAX phenotypes that is derived directly from the univariate GWAS and GWAX estimates. The associated V matrix is also expanded using cross-trait

intercepts from LDSC in order to take any potential sample overlap (known or unknown) and/or shared stratification implied by the LDSC model into account. In the second stage, the model is fit to the S matrix, and free parameters are estimated such that they minimize the discrepancy between Σ and S using the diagonally weighted least squares (WLS) fit function with sandwich correction, with weights derived from V, as described in Grotzinger et al. [18].

In a sensitivity analysis of the empirical data, we estimated the S matrix and its associated V matrix using stratified LDSC instead of LDSC, so as to allow for potentially uneven contributions of variants to heritability across biologically, evolutionarily, and MAF defined categories. Stratified LDSC was estimated with stratified Genomic SEM using 97 baseline annotations [21,32].

Note that the liability scale heritability is sensitive to assumptions about the population prevalence of AD. In our analyses of empirical data, our primary models set the population prevalence rate of AD at 5% used by Wightman et al. [4] to produce their estimate of liability-scale h² of 2.5%., but we consider sensitivity of liability scale heritability estimates to other values, and to the possibility of undiagnosed AD within control participants (see Prevalence Rates in Relation to the Heritability of Clinical AD and AD Pathology, below). We also specifically compare estimates to those obtained with a population prevalence of 5% used by Wightman et al. [4] to produce their estimate of liability-scale h² of 2.5%.

Simulation study of SNP heritability

Simulation of summary statistics. We simulated genome-wide summary statistics for direct GWAS and maternal and paternal GWAX manipulating the attenuation coefficients (λ) for maternal and paternal GWAX assuming both continuous and binary traits. For binary traits we matched the sample prevalence for direct GWAS and maternal and paternal GWAX with those from IGAP ($\nu = 0.344$), UKB maternal ($\nu = 0.096$) and UKB paternal ($\nu = 0.055$), and a population prevalence of .05. S1 Table provides the details of the experimental design of our simulation study. For continuous traits, we provide six supplemental experimental conditions in which we additionally manipulated the sample size for the GWAX phenotypes compared with the direct GWAS (S10 Table). We simulated data for three phenotypes (direct GWAS, maternal GWAX, and paternal GWAX) per replication, and 100 replications per each of the conditions, for a total of 3,600 simulated sets of summary statistics. (As described below, each set of three summary statistics is then analyzed according to three meta-analytic methods, for a total of 3,600 meta-analyses). We used the parameter specifications provided in Table 1 to produce implied population-level genetic covariance matrices, Σ , calculated as

$$\begin{bmatrix} h_1^2 & & \\ \sigma_{g1,2} & h_2^2 & \\ \sigma_{g1,3} & \sigma_{g2,3} & h_3^2 \end{bmatrix} = \Lambda \Psi \Lambda' = \begin{bmatrix} \lambda_1^2 h_F^2 & & \\ \lambda_1 \lambda_2 h_F^2 & \lambda_2^2 h_F^2 & \\ \lambda_1 \lambda_3 h_F^2 & \lambda_2 \lambda_3 h_F^2 & \lambda_3^2 h_F^2 \end{bmatrix}$$

Where $\Psi = h_F^2$ and Λ is the vector of attenuation coefficients (i.e. factor loadings, λ). We then used the LD scores for European population for M = 1,184,461 common Hapmap3 SNPs (MAF>.01 excluding the MHC region) provided by Bulik-Sullivan et al. [33], to simulate summary data for three phenotypes according to the multivariate LDSC equation, i.e.

$$[Z_{1j}, Z_{2j}, Z_{3j}] \sim \mathsf{N}([0, 0, 0], \operatorname{cov}(Z_{1j}, Z_{2j}, Z_{3j})),$$

where

$$\begin{split} & \operatorname{cov}(Z_{1j}, Z_{2j}, Z_{3j}) \\ & = \begin{bmatrix} N_1 \frac{h_1^2}{M} \ell(j) + 1 + a_1 \\ & \sqrt{N_1 N_2} \frac{\sigma_{g1,2}}{M} \ell(j) + \frac{\rho_{12} N_{s12}}{\sqrt{N_1 N_2}} & N_2 \frac{h_2^2}{M} \ell(j) + 1 + a_2 \\ & \sqrt{N_1 N_3} \frac{\sigma_{g1,3}}{M} \ell(j) + \frac{\rho_{13} N_{s13}}{\sqrt{N_1 N_3}} & \sqrt{N_2 N_3} \frac{\sigma_{g23}}{M} \ell(j) + \frac{\rho_{23} N_{s23}}{\sqrt{N_2 N_3}} & N_3 \frac{h_3^2}{M} \ell(j) + 1 + a_3 \end{bmatrix} \end{split}$$

and $[Z_{1j}, Z_{2j}, Z_{3j}]$ represents the Z statistics for the three GWAS/GWAX phenotypes, N is the sample size of the corresponding GWAS/GWAX, M is the number of SNPs including in the LD file, $\ell(j)$ is the LD score of SNP j (that is, the sum of squared correlations between the SNP and all other SNPs), N_s is the number of overlapping individuals in the corresponding GWAS/ GWAX, ρ is the phenotypic correlation within the overlapping individuals, and α represents unmeasured sources of confounding such as population stratification.

We additionally simulated GWAS and GWAX summary data from a null scenario in which the model-implied genetic covariance matrix consisted of a diagonal matrix with heritabilities for direct, maternal, and paternal traits equal to .10, .025, and .025, respectively. In this scenario, genetic architecture is not shared across the GWAS and GWAX, and the multivariate heritability estimate is therefore expected to be 0. We fitted our multivariate Genomic SEM relaxed model on the simulated summary statistics to quantify Type I error rates for the model estimate of h² across 100 replications.

Our simulation approach, in which summary statistics are directly generated from the LDSC equation is particularly well-suited to our purposes. Direct generation of summary statistics allows us to consider an expansive set of replications and conditions (i.e. 3,600 sets of simulated summary statistics total) that would be computationally prohibitive to simulate using a framework in which raw phenotype data were first generated for individual genomes and then submitted to GWAS. Summary data simulated under the LDSC model has the properties needed for analysis of genetic architecture by LDSC and for meta-analysis of effect sizes across phenotypes on a per-variant basis. Importantly, like any simulation approach, our approach also lacks some nuances of real data. For example, while we simulate summary data as a function of LD scores, we do not simulate summary data directly according to linkage disequilibrium (LD) structure (this would require the expansion of the covariance matrix of Z statistics across phenotypes to include cross-SNP covariances, which would dramatically increase computational burden). Thus, the simulated summary data are not appropriate for applications that are directly based on LD structure such as clumping and pruning, identification of lead SNPs within loci, or estimation of heritability using methods such as HDL [34] and HESS [23], which directly rely on LD structure (rather than simply on LD scores).

We present results of simulations in which we generate summary statistics under conditions of no sample overlap ($N_s = 0$) and no population stratification (a = 0). However, within this analytic framework, and in previous work using raw data simulation [18] we have confirmed that Genomic SEM produces unbiased estimates and standard errors when summary statistics are generated with sample overlap, phenotypic correlation within overlapping individuals, and population stratification.

Analysis of simulated summary statistics

We compared the performance of three approaches to recover common variant observed SNP heritability for continuous traits and liability-scale heritability for binary traits: 1) LDSC of summary statistics from a conventional meta-analysis of Z statistics using the observed sample sizes ("Conventional Approach [Uncorrected]"), 2) LDSC of summary statistics from a conventional inverse-variance weighted meta-analysis of corrected betas and SEs assuming a simple GWAX model ("Conventional Approach [Standard Correction]"), and 3) our multivariate Genomic SEM-based approach, using the minimal identification constraint $\lambda_{direct} = 1$ such that $\sigma_{\rm F}^2$ corresponds to the estimate of $h_{\rm F}^2$). For binary traits we additionally introduced a correction for the conversion from observed to liability scale heritability ("Conventional Approach [Standard + Liability Correction]"). The liability correction consisted of using the sum of effective sample sizes (Neff, defined as $4v_k(1-v_k)n_k$, where v_k is the sample prevalence for study k) multiplied by the expected heritability attenuation based on a simple GWAX model (i.e., Neff_{direct}+.25Neff_{mat}+.25Neff_{pat}). Because Neff represents the sample size for an equally balanced GWAS in terms of % of cases and controls (i.e., 50% cases, 50% controls), the sum of effective sample sizes is then entered in the LDSC stage, along with a .5 as sample prevalence. See Grotzinger et al. [31] for a more detailed explanation on the liability-scale correction based on the sum of effective sample sizes. The conventional uncorrected approach is reflective of the approaches taken by Jansen et al. [16] and Wightman et al. [4]. The conventional corrected approach is reflective of the approaches taken by Marioni et al. [3] and Bellenguez et al. [5]. The conventional approach with standard and liability correction combines the approaches taken by Marioni et al. [3] and Bellenguez et al. [5] with the liability-scale correction introduced by Grotzinger et al. [31], thus maintaining the standard GWAX assumptions while eliminating bias arising from differences in ascertainment across the GWAS and GWAXs.

For each condition, we calculated: 1) mean parameter estimate, 2) standard deviation (SD) of the parameter estimate, 3) mean standard error (SE) of the parameter estimate, and 4) Percent Bias Error (%BE) of the parameter estimate, calculated as:

$$\text{\%BE} = \frac{1}{100} \sum_{r=1}^{100} \left(\frac{h_{l,r}^2}{h_{l,True}^2} - 1 \right)$$

Where $h_l^2 r$ is the parameter estimate in replication *r*, and $h_{l,True}^2$ is its true value for that particular condition.

Simulation study of individual SNP effects

We extended our simulations for recovery of heritability to allow us to test for the recovery of individual SNP effects for both continuous and binary traits. Although the simulation for recovery of heritability generated individual level SNP effects, those analyses simulated based on a population model with a known genetic covariance structure, but not known effects for individual SNPs. However, in order to benchmark performance of each method for estimating individual SNP effects we must know their true effect size in the population from which we draw our simulation. We therefore extended the simulation analyses to include a single individual SNP per replication with a known true effect according to the conditions in S1 Table. We set the true effect (γ) of this SNP on *F* to be equal to the partially standardized (i.e. standard deviations in liability for AD per effect allele) beta coefficient of genome-wide-significant SNP rs60738304 from the IGAP summary statistics (*b* = .068, MAF = 0.305), such that

$$[b_{1j}, b_{2j}, b_{3j}] = \gamma \Lambda = [.068\lambda_1, .068\lambda_2, .068\lambda_3].$$

For continuous traits, we sampled observed regression coefficients from their sample-size dependent sampling distribution as follows

For binary traits, the regression coefficients were sampled from their sampling distribution as:

$$\begin{bmatrix} \hat{b}_1, \hat{b}_2, \hat{b}_3 \end{bmatrix} \sim \mathbf{N} \left(\begin{bmatrix} b_1, b_2, b_3 \end{bmatrix}, \begin{bmatrix} \frac{1 - 2MAF(1 - MAF)b_1^{-2}}{2MAF(1 - MAF)\nu(1 - \nu)n_1} \\ 0 & \frac{1 - 2MAF(1 - MAF)b_2^{-2}}{2MAF(1 - MAF)\nu(1 - \nu)n_2} \\ 0 & 0 & \frac{1 - 2MAF(1 - MAF)b_3^{-2}}{2MAF(1 - MAF)\nu(1 - \nu)n_3} \end{bmatrix} \right)$$

Note that because we simulate under conditions of no sample overlap across any of the direct GWAS and two GWAX datasets, the sampling covariance matrix of the coefficients is diagonal in the population model (the corresponding cells in the V matrix are nevertheless freely estimated when the data are analyzed in Genomic SEM, such that any sample overlap that does exist is automatically detected and corrected for).

Analysis of simulated data

We computed meta-analytic estimates of γ using the conventional corrected and uncorrected approaches, and the multivariate Genomic SEM approach (model with individual SNP effects using the minimal identification constraint $\lambda_{direct} = 1$, such that γ takes on the scale of the direct GWAS). Because the Genomic SEM approach requires a genetic covariance matrix that includes terms for individual SNP effects and terms for genome-wide genetic covariances (the genome-wide covariance structure informs the estimation of individual SNP effects), we converted the simulated regression coefficients to SNP-level genetic covariances and appended them to genome-wide genetic covariance matrices estimated from LDSC using the genome-wide summary statistics produced under the simulation of SNP heritability for the corresponding condition.

For each simulation condition, we calculated: 1) mean parameter estimate, 2) standard deviation (SD) of the parameter estimate, 3) mean standard error (SE) of the parameter estimate, 4) mean Z statistic of the parameter estimate, and 5) Percent Bias Error (%BE) of the parameter estimate.

Selection of GWAS and GWAX summary data

We compiled summary data from three published European ancestry direct case-control GWAS of AD and GWAX of parental history of AD.

Direct case-control GWAS summary statistics encompassed the discovery sample from the IGAP consortium [12] comprising 21,982 Clinical AD cases (mean age of onset = 72.93 years) and 41,944 controls (mean age of evaluation = 72.415 years). Cohorts contributing to IGAP varied considerably in the extent to which clinical determination of case and control status was confirmed by autopsy. To avoid bias otherwise arising from variation in case prevalence across cohorts contributing to IGAP, we followed the approach described by Grotzinger et al. [31], using the sum of effective sample sizes $(4v_k(1-v_k)n_k)$, where v_k is the sample prevalence for each GWAS contributing to the GWAS meta-analysis). Our GWAX summary data of proxy-phenotype AD included 27,696 cases and 260,980 controls of history of maternal AD, and 14,338 cases and 245,941 controls of history of paternal AD, both phenotypes from UK Biobank [3]. Case status for UK Biobank was determined by response to the following two questions "Has/ did your father ever suffer from Alzheimer's disease/dementia?" and "Has/did your mother ever suffer from Alzheimer's disease/dementia?" at the initial assessment visit (2006-2010), the first repeat assessment visit (2012–2013) and the imaging visit (2014+). Participants whose parents were younger than 60 years or died prior to age 60 years, and without parental age information were excluded. Zhang et al. [28] report that the mean age of maternal cases was 83.7 years, the mean age of paternal cases was 81.8 years, the mean age of maternal controls was 78.1 years, and the mean age of paternal controls was 76.2 years. Further details on case ascertainment, genotyping, and quality control can be found in the original articles for the corresponding summary statistics.

We additionally curated the following GWAS summary statistics for the estimation of genetic correlations: two recent meta-analyses of direct GWAS and GWAX of AD [3,16], brain volume [35] in the general population, educational attainment [36] in the general population and a general genetic factor of cognitive function [37] in the general population.

Quality control of GWAS and GWAX summary aata

LD-Score Regression (LDSC). All summary statistics used for LDSC were cleaned and processed using the munge function of Genomic SEM, using the standard practice of retaining all HapMap3 [38] SNPs outside of the major histocompatibility complex (MHC) region with minor allele frequencies (MAFs) \geq .01 and information scores (INFO) > .9. To avoid misfit in LD score regression due to extremely large effect sizes within the *APOE* region, we additionally remove this region (CHR19:45,116,911–46,318,605). This has a similar effect to the standard practice in LDSC of removing SNPs with extremely large test statistics. The LD scores used for the analyses presented were estimated from the European sample of 1000 Genomes, but restricted to HapMap3 SNPs as these tend to be well-imputed and produce accurate estimates of heritability. In the LDSC equation, we enter the total number of SNPs in the reference LD panel before excluding the MHC and *APOE* regions, such that the SNP heritability estimates reported here retain similar interpretation to those typically produced by LDSC in other contexts. Based on these procedures, the SNP heritability estimates produced by LDSC should be considered estimates of heritability explained by common SNPs, not including the extremely strong effect of *APOE*.

Multivariate GWAS

For the SNP portion of the multivariate model that included individual SNP effects, we used the default QC procedures in Genomic SEM [18] of removing SNPs with an MAF < .005 in the 1000 Genomes Phase 3 reference panel and SNPs with an INFO score < 0.6 in the

univariate GWAS summary statistics. Using these QC procedures left 7,192,577 SNPs across the three contributing summary statistics datasets. Prior to running any multivariate GWAS, all SNP effects were converted to logistic regression coefficients, standardized with respect to the total liability scale variance in the outcome using the sumstats function in GenomicSEM, and corrected for genomic inflation by multiplying the standard errors by square root of the univariate LDSC intercept when the intercept was above 1. These transformed estimates were then multiplied by the SNP variance (estimated as 2*MAF(1-MAF)) to produce genetic covariances. In the case of some variants within the *APOE* region with very large effects, we reduced the MAF relative to its reference panel value when calculating the SNP variance in order to prevent the full S matrix (containing both SNP effects and genetic covariances among the direct GWAS and two GWAX phenotypes) from being nonpositive definite. As we ultimately estimate SNP effects in per effect-allele units, rather than in per standard deviation units, this decision does not bias the estimates of interest.

Post GWAS analysis of meta-analytic results in FUMA

We applied FUMA [39] to results of our multivariate GWAS meta-analysis to identify independent significant SNPs, lead SNPs, and risk loci, using the defaults. Independent significant SNPs were defined as genome wide significant SNPs that are independent from one another at r^2 <.60. Independent significant SNPs are clumped to identify lead SNPs, which are independent from each other at r^2 < 0.1. In defining genomic risk loci, independent significant SNPs that are associated $r^2 \ge .10$ are assigned to the same locus, and independent significant SNPs closer than 250kb are merged into a single locus. The most significant lead SNP in a locus is used to represent that locus. The EUR population from 1000 Genomes Phase 3 was used as the reference panel. As 16 SNPs in the *APOE* block (rs429358) produced very large Z statistics such that their p-value was treated by FUMA as 0 and excluded, we manually entered the most significant SNP in the *APOE* block into FUMA as the lead SNP.

Prevalence rates in relation to the heritability of clinical AD and biological AD

Clinical AD. When methods for estimating SNP heritability, such as LDSC, are applied to GWAS of binary phenotypes, the estimate of heritability are on the *observed scale*. Observed scale heritability is difficult to interpret, because it takes on a metric that is idiosyncratic to the balance of cases and controls in the sample, and agnostic to the population prevalence rates of the categories. It is more interpretable to convert observed scale heritability to liability scale heritability based on a liability threshold model, which assumes that a normally distributed liability toward the binary phenotype exists in a population. Under this model, individuals with a liability above a certain threshold are affected whereas those below this threshold are unaffected. The liability-scale SNP heritability estimate is therefore an estimate of the proportion of variation in this continuous liability that is explained by all (directly and indirectly tagged) SNPs on which the GWAS is based. For ascertained samples, in which the proportion of cases does not represent their prevalence in the population, and assuming that the control participants are successfully screened, such that they are unaffected by the disorder (an issue that we return to next), the expected liability scale heritability (\hat{h}_i^2) can be obtained from observed scale heritability (h_i^2) according to Peyrot et al. [40]

$$\hat{h}_l^2 = rac{P_{population}^2 (1-P_{population})^2}{P_{sample}(1-P_{sample}) arphi(t)^2} h_o^2,$$

where P_{sample} is the proportion of cases in the GWAS sample, $P_{population}$ is the proportion of cases in the population (i.e. the population prevalence), *t* is the threshold of the cumulative normal distribution corresponding to the population prevalence, and $\varphi(t)$ is the density of the normal distribution at *t*.

Whereas P_{sample} is known, $P_{population}$ must be obtained from collateral epidemiological data. As Yang et al. [41] write, "Application of [the transformation from observed scale heritability to liability scale heritability] requires an estimate of [the population prevalence of the disorder], for which estimates can be surprisingly hard to find, and most applications include a sensitivity analysis to the choice of [this parameter]." Following this guidance, we provide estimates of liability scale heritability across a range of different prevalence rates, and provide rough age equivalents of these prevalence rates based on published epidemiological data from the Medical Research Council Cognitive Function and Ageing Study II [19]. We refer to this estimate as the estimate of heritability of Clinical AD, in that this is the estimate based on the prevalence rate of the clinical diagnosis, and does not account for undetected biological AD that may be more prevalent in the population.

Biological AD

The past decade of research on AD has been increasingly cognizant of the fact that "the pathophysiological process of [AD] begins years, if not decades, before the diagnosis of dementia" [42]. Those presenting with Clinical AD may not constitute those individuals with the most severe forms of AD pathophysiology, but may be constitute a group whose cognitive impairments are more overt or detectable due to ancillary characteristics (such as those related to peak premorbid levels of cognitive functioning) that are unrelated to the genetics of biological AD [27]. Indeed, this common observation that differences in ancillary factors such as educational attainment are related to differences in Clinical AD rates among individuals with equivalent levels of brain pathology has led to the popular concept of *cognitive reserve* [7,20,43]. In order to estimate the heritability of biological AD, we capitalize on the Clinical AD GWAS and the age-specific rates of biological AD. We employ the formula for converting observed to liability scale heritability that corrects for case contamination in control participants provided by Peyrot et al. [40], further adapting to allow for imperfect screening of participants (i.e. screening for Clinical but not biological AD):

$$\hat{h}_{l}^{2} = \frac{P_{\text{population Biological AD}}^{2} (1 - P_{\text{population Biological AD}})^{2}}{P_{\text{sample Clinical AD}} (1 - P_{\text{sample Clinical AD}}) (1 - FCR)^{2} \varphi(t)^{2}} h_{o}^{2}$$

where *FCR* is the false classification rate: the proportion of incorrectly classified control participants in the GWAS. When control participants are a random sample of the population, we set $FCR = P_{population}$. However, because controls have been screened for clinical AD but not biological AD, and we estimate the liability scale heritability for biological AD under the assumption that clinical AD represents a random subset of biological AD, and compute *FCR* as

$$FCR = (P_{population Biological AD} - P_{population Clinical AD})$$

We provide estimates of liability scale heritability of biological AD according to this equation across a range of different population prevalence rates for biological AD, and provide rough age equivalents of these prevalence rates based on recently published data from Jack et al. [10]. To achieve this, we fix $P_{population \ Clinical \ AD}$ at the approximate age-specific estimate of the clinical AD prevalence rate from epidemiological data for the mean age of the control sample in IGAP (72.42 years, prevalence rate = 0.0285) to reflect the expected proportion of participants that have been screened out of the control sample for clinical AD) and we vary $P_{population Biological AD}$.

Local SNP heritability in HESS

We used HESS to estimate local common variant SNP heritability in 1,703 approximately independent blocks across the genome, using common variants (MAF>.01) in the EUR population from 1000 Genomes as the reference panel [23]. Because LDSC relies on the variation in average LD across SNPs to estimate SNP heritability, it cannot be used to estimate local heritability within individual loci in which LD is relatively homogeneously high. In contrast, because HESS relies on the LD matrix itself to estimate SNP heritability, and does not specifically rely on variation in average LD, it is more appropriate for estimating local heritability.

Supporting information

S1 Supplementary Note. Integrated Analysis of Direct and Proxy Genome Wide Association Studies Highlights Polygenicity of Alzheimer's Disease outside of the APOE Region. (DOCX)

S2 Supplementary Results. Integrated Analysis of Direct and Proxy Genome Wide Association Studies Highlights Polygenicity of Alzheimer's Disease outside of the APOE Region. (DOCX)

S1 Fig. Path diagram for multivariate genetic analysis of Alzheimer's disease. We present the minimal identification constraint $\lambda_{direct} = 1$ such that the variance of the factor corresponds to the meta-analytic heritability estimate on scale of the direct GWAS. The dashed portion of the diagram represents the portion of the model that is specified to produce meta-analytic estimates for effects of individual SNPs. (TIF)

S2 Fig. Simulation Results. Distribution of observed SNP heritability estimates (h_{SNP}^2) across supplemental conditions S1-S6 for continuous traits. (TIFF)

S3 Fig. Simulation Results. Distribution of individual SNP effects across supplemental conditions S1-S6 for continuous traits. (TIFF)

S4 Fig. Heatmaps of LDSC cross-trait intercepts (left) and genetic correlations (right) among direct GWAS, maternal and paternal GWAX, and meta-analytic summary statistics of Alzheimer's disease from two previous studies, and from the Genomic SEM-based multivariate method introduced here.

(TIFF)

S5 Fig. Manhattan plot for the common factor GWAS of Alzheimer's disease. *Note:* the upper limit for the y axis was constrained to 50, the APOE region goes off-scale (lead SNP rs429358 $-\log_{10}(p) = 308.653$). (TIFF)

S6 Fig. QQ-plot for AD Genomic SEM relaxed model and heterogeneity Q. Mean χ^2 for AD GSEM = 1.139. Mean χ^2 for heterogeneity Q = 1.008. (TIF)

S7 Fig. Estimated common variant SNP heritability of AD (outside of the MHC and APOE regions) according to different assumptions regarding the population prevalence of AD and biological AD, based on LDSC of summary statistics from our multivariate meta-analysis. We provide rough approximations of the age equivalences of each prevalence rate on the top x axis. The purple diamond represents the estimate of 2.5% by Wightman et al.[4], which was based on an assumed population prevalence rate of 5%. The yellow diamond represents the estimate from the multivariate model introduced here, using the same assumed population prevalence rate of 5% (Clinical AD $h^2 = 0.069$; Biological AD $h^2 = 0.072$). The shading area around the line reflects +/-1 SE of the h² estimate. The steeper shift in the SNP heritability of AD for biological AD compared to clinical AD as a function of population prevalence stems from the correction for undetected biological AD within control participants who primarily only been screened for clinical AD. Thus, as the assumed prevalence rate of biological AD increases, the extent of case contamination in control participants increases, and the correction for undetected AD in control participants produces more dramatic increases in the projected heritability. (TIFF)

S8 Fig. Association between local SNP heritability and GWAS effect sizes for GWAS loci that were genome wide significant in the multivariate GWAS including the APOE locus containing lead SNP rs429358 (left) and excluding this locus (right).

(TIFF)

S9 Fig. Local SNP heritability results for IGAP. (TIFF)

S10 Fig. Heatmap of LDSC genetic correlations among meta-analyses of case-control and proxy-phenotype family history of Alzheimer's disease and external correlates of Alzheimer's disease.

(TIFF)

S1 Table. Experimental conditions for the simulation of summary statistics and SNP effect sizes for binary traits.

(XLSX)

S2 Table. Recovery of SNP-based heritability across experimental conditions and methods (100 runs per condition). (XLSX)

(ALSA)

S3 Table. Recovery of individual SNP effects across experimental conditions and methods (100 runs per condition).

(XLSX)

S4 Table. Genome-wide significant (P < 5 x 10–8) SNP association results for Genomic SEM AD.

(XLSX)

S5 Table. Genome-wide significant risk loci associated with Genomic SEM AD. (XLSX)

S6 Table. Effect sizes associated with lead SNPs of genomic risk loci across contributing datasets and meta-analytic approaches. (XLSX) S7 Table. Lead SNPs and tagged SNPs for GSEM AD previously identified in GWAS of diseases and traits listed in the GWAS catalog.

(XLSX)

S8 Table. Independent lead SNPs and SNPs tagged within each significant genomic locus for GSEM AD.

(XLSX)

S9 Table. Type I error rates for heritability estimates simulated from a null model (100 iterations).

(XLSX)

S10 Table. Supplemental experimental conditions for the simulation of summary statistics and SNP effect sizes for continuous traits.

(XLSX)

S11 Table. Recovery of SNP-based heritability across supplemental experimental conditions and methods for continuous traits (100 runs per condition). (XLSX)

S12 Table. Recovery of individual SNP effects across supplemental experimental conditions and methods for continuous traits (100 runs per condition). (XLSX)

Author Contributions

Conceptualization: Javier de la Fuente, Elliot M. Tucker-Drob.

Data curation: Javier de la Fuente.

Formal analysis: Javier de la Fuente, Andrew D. Grotzinger, Elliot M. Tucker-Drob.

Methodology: Javier de la Fuente, Elliot M. Tucker-Drob.

Resources: Riccardo E. Marioni.

Software: Javier de la Fuente, Andrew D. Grotzinger.

Supervision: Elliot M. Tucker-Drob.

Visualization: Javier de la Fuente.

Writing - original draft: Javier de la Fuente, Elliot M. Tucker-Drob.

Writing – review & editing: Javier de la Fuente, Andrew D. Grotzinger, Riccardo E. Marioni, Michel G. Nivard, Elliot M. Tucker-Drob.

References

- Liu JZ, Erlich Y, Pickrell JK. Case-control association mapping by proxy using family history of disease. Nat Genet. 2017; 49(3):325. https://doi.org/10.1038/ng.3766 PMID: 28092683
- Hujoel MLA, Gazal S, Loh P-R, Patterson N, Price AL. Liability threshold modeling of case-control status and family history of disease increases association power. Nat Genet. 2020; 52(5):541–7. https://doi.org/10.1038/s41588-020-0613-6 PMID: 32313248
- Marioni RE, Harris SE, Zhang Q, McRae AF, Hagenaars SP, Hill WD, et al. GWAS on family history of Alzheimer's disease. Transl Psychiatry. 2018; https://doi.org/10.1038/s41398-018-0150-6 PMID: 29777097
- Wightman DP, Jansen IE, Savage JE, Shadrin AA, Bahrami S, Rongve A, et al. Largest GWAS (N = 1,126,563) of Alzheimer's Disease Implicates Microglia and Immune Cells. medRxiv. 2020;

- Bellenguez C, Küçükali F, Jansen I, Andrade V, Moreno-Grau S, Amin N, et al. New insights on the genetic etiology of Alzheimer's and related dementia. medRxiv. 2020;
- 6. APA (American Psychiatric Association). Diagnostic and Statistical Manual of Mental Disorders, 5th Edition: DSM-5. Washington, DC: APA.; 2013.
- Tucker-Drob EM. Cognitive Aging and Dementia: A Life-Span Perspective. Annu Rev Dev Psychol. 2019; 1(1):177–96. https://doi.org/10.1146/annurev-devpsych-121318-085204 PMID: 34046638
- McKhann GM, Knopman DS, Chertkow H, Hyman BT, Jack CR Jr., Kawas CH, et al. The diagnosis of dementia due to Alzheimer's disease: Recommendations from the National Institute on Aging-Alzheimer's Association workgroups on diagnostic guidelines for Alzheimer's disease. Alzheimer's Dement [Internet]. 2011 May 1; 7(3):263–9. Available from: https://doi.org/10.1016/j.jalz.2011.03.005
- Jack CR, Bennett DA, Blennow K, Carrillo MC, Dunn B, Haeberlein SB, et al. NIA-AA Research Framework: Toward a biological definition of Alzheimer's disease. Alzheimer's Dement [Internet]. 2018; 14 (4):535–62. Available from: https://doi.org/10.1016/j.jalz.2018.02.018
- Jack CR, Therneau TM, Weigand SD, Wiste HJ, Knopman DS, Vemuri P, et al. Prevalence of Biologically vs Clinically Defined Alzheimer Spectrum Entities Using the National Institute on Aging-Alzheimer's Association Research Framework. JAMA Neurol. 2019; 76(10):1174–83. <u>https://doi.org/10.1001/jamaneurol.2019.1971</u> PMID: 31305929
- Gatz M, Reynolds CA, Fratiglioni L, Johansson B, Mortimer JA, Berg S, et al. Role of genes and environments for explaining Alzheimer disease. Arch Gen Psychiatry. 2006; 63(2):168–74. https://doi.org/ 10.1001/archpsyc.63.2.168 PMID: 16461860
- Kunkle BW, Grenier-Boley B, Sims R, Bis JC, Damotte V, Naj AC, et al. Genetic meta-analysis of diagnosed Alzheimer's disease identifies new risk loci and implicates Aβ, tau, immunity and lipid processing. Nat Genet. 2019; 51(3):414–30. https://doi.org/10.1038/s41588-019-0358-2 PMID: 30820047
- Anttila V, Bulik-Sullivan B, Finucane HK, Walters RK, Bras J, Duncan L, et al. Analysis of shared heritability in common disorders of the brain. Science (80-). 2018; https://doi.org/10.1126/science.aap8757 PMID: 29930110
- Lee SH, Harold D, Nyholt DR, Goddard ME, Zondervan KT, Williams J, et al. Estimation and partitioning of polygenic variation captured by common SNPs for Alzheimer's disease, multiple sclerosis and endometriosis. Hum Mol Genet. 2013; 22(4):832–41. https://doi.org/10.1093/hmg/dds491 PMID: 23193196
- Sherva R, Gross A, Mukherjee S, Koesterer R, Amouyel P, Bellenguez C, et al. Genome-wide association study of rate of cognitive decline in Alzheimer's disease patients identifies novel genes and pathways. Alzheimer's Dement. 2020; 16(8):1134–45. https://doi.org/10.1002/alz.12106 PMID: 32573913
- Jansen IE, Savage JE, Watanabe K, Bryois J, Williams DM, Steinberg S, et al. Genome-wide metaanalysis identifies new loci and functional pathways influencing Alzheimer's disease risk. Nat Genet. 2019; 51(3):404–13. https://doi.org/10.1038/s41588-018-0311-9 PMID: 30617256
- Turley P, Walters RK, Maghzian O, Okbay A, Lee JJ, Fontana MA, et al. Multi-trait analysis of genomewide association summary statistics using MTAG. Nat Genet. 2018; <u>https://doi.org/10.1038/s41588-017-0009-4</u> PMID: 29292387
- Grotzinger AD, Rhemtulla M, de Vlaming R, Ritchie SJ, Mallard TT, Hill WD, et al. Genomic structural equation modelling provides insights into the multivariate genetic architecture of complex traits. Nat Hum Behav. 2019; https://doi.org/10.1038/s41562-019-0566-x PMID: 30962613
- Matthews FE, Arthur A, Barnes LE, Bond J, Jagger C, Robinson L, et al. A two-decade comparison of prevalence of dementia in individuals aged 65 years and older from three geographical areas of England: Results of the cognitive function and ageing study i and II. Lancet [Internet]. 2013; 382 (9902):1405–12. Available from: https://doi.org/10.1016/S0140-6736(13)61570-6 PMID: 23871492
- 20. Stern Y. Cognitive reserve in ageing and Alzheimer's disease. Lancet Neurol. 2012; 11(11):1006–12. https://doi.org/10.1016/S1474-4422(12)70191-6 PMID: 23079557
- Grotzinger AD, Mallard TT, Akingbuwa WA, Ip HF, Adams MJ, Lewis CM, et al. Genetic Architecture of 11 Major Psychiatric Disorders at Biobehavioral, Functional Genomic, and Molecular Genetic Levels of Analysis. medRxiv [Internet]. 2020 Jan 1;2020.09.22.20196089. Available from: http://medrxiv.org/ content/early/2020/09/23/2020.09.22.20196089.abstract
- Finucane HK, Bulik-Sullivan B, Gusev A, Trynka G, Reshef Y, Loh PR, et al. Partitioning heritability by functional annotation using genome-wide association summary statistics. Nat Genet. 2015 Nov; 47 (11):1228–35. https://doi.org/10.1038/ng.3404 PMID: 26414678
- Shi H, Kichaev G, Pasaniuc B. Contrasting the genetic architecture of 30 complex traits from summary association data. Am J Hum Genet. 2016; 99(1):139–53. <u>https://doi.org/10.1016/j.ajhg.2016.05.013</u> PMID: 27346688

- Sinnott-Armstrong N, Naqvi S, Rivas M, Pritchard JK. GWAS of three molecular traits highlights core genes and pathways alongside a highly polygenic background. Elife. 2021; 10:e58615. <u>https://doi.org/</u> 10.7554/eLife.58615 PMID: 33587031
- Schwartzentruber J, Cooper S, Liu JZ, Barrio-Hernandez I, Bello E, Kumasaka N, et al. Genome-wide meta-analysis, fine-mapping and integrative prioritization implicate new Alzheimer's disease risk genes. Nat Genet. 2021; 53(3):392–402. https://doi.org/10.1038/s41588-020-00776-w PMID: 33589840
- Boyle EA, Li YI, Pritchard JK. An Expanded View of Complex Traits: From Polygenic to Omnigenic. Cell. 2017 Jun; 169(7):1177–86. https://doi.org/10.1016/j.cell.2017.05.038 PMID: 28622505
- 27. Lövdén M, Fratiglioni L, Glymour MM, Lindenberger U, Tucker-Drob EM. Education and Cognitive Functioning Across the Life Span. Psychol Sci Public Interes. 2020; 21(1):6–41. https://doi.org/10. 1177/1529100620920576 PMID: 32772803
- Zhang Q, Sidorenko J, Couvy-Duchesne B, Marioni RE, Wright MJ, Goate AM, et al. Risk prediction of late-onset Alzheimer's disease implies an oligogenic architecture. Nat Commun. 2020; 11(1):1–11.
- Bellenguez C, Küçükali F, Jansen I, Andrade V, Morenau-Grau S, Amin N, et al. Large meta-analysis of genome-wide association studies expands knowledge of the genetic etiology of Alzheimer's disease and highlights potential translational opportunities. MedRxiv. 2020;
- Wightman DP, Jansen IE, Savage JE, Shadrin AA, Bahrami S, Holland D, et al. A genome-wide association study with 1,126,563 individuals identifies new risk loci for Alzheimer's disease. Nat Genet. 2021; 53(9):1276–82. https://doi.org/10.1038/s41588-021-00921-z PMID: 34493870
- Grotzinger AD, de la Fuente J, Nivard MG, Tucker-Drob EM. Pervasive Downward Bias in Estimates of Liability Scale Heritability in GWAS Meta-Analysis: A Simple Solution. medRxiv. 2021 Jan:2021.09.22.21263909.
- **32.** Grotzinger AD, de la Fuente J, Davies G, Nivard MG, Tucker-Drob EM. Transcriptome-wide and Stratified Genomic Structural Equation Modeling Identify Neurobiological Pathways Underlying General and Specific Cognitive Functions. medRxiv. 2021 Jan;2021.04.30.21256409.
- Bulik-Sullivan B, Loh PR, Finucane HK, Ripke S, Yang J, Patterson N, et al. LD score regression distinguishes confounding from polygenicity in genome-wide association studies. Nat Genet. 2015; 47 (3):291–5. https://doi.org/10.1038/ng.3211 PMID: 25642630
- Ning Z, Pawitan Y, Shen X. High-definition likelihood inference of genetic correlations across human complex traits. Nat Genet. 2020; 52(8):859–64. <u>https://doi.org/10.1038/s41588-020-0653-y</u> PMID: 32601477
- Zhao B, Zhang J, Ibrahim JG, Luo T, Santelli RC, Li Y, et al. Large-scale GWAS reveals genetic architecture of brain white matter microstructure and genetic overlap with cognitive and mental health traits (n = 17,706). Mol Psychiatry. 2019;
- Lee JJ, Wedow R, Okbay A, Kong E, Maghzian O, Zacher M, et al. Gene discovery and polygenic prediction from a genome-wide association study of educational attainment in 1.1 million individuals. Nat Genet. 2018; 50(8):1112–21. https://doi.org/10.1038/s41588-018-0147-3 PMID: 30038396
- de la Fuente J, Davies G, Grotzinger AD, Tucker-Drob EM, Deary IJ. A general dimension of genetic sharing across diverse cognitive traits inferred from molecular data. Nat Hum Behav. 2020; <u>https://doi.org/10.1038/s41562-020-00936-2</u> PMID: 32895543
- Consortium IH 3. Integrating common and rare genetic variation in diverse human populations. Nature. 2010; 467(7311):52. https://doi.org/10.1038/nature09298 PMID: 20811451
- Watanabe K, Taskesen E, Van Bochoven A, Posthuma D. Functional mapping and annotation of genetic associations with FUMA. Nat Commun. 2017; https://doi.org/10.1038/s41467-017-01261-5 PMID: 29184056
- Peyrot WJ, Boomsma DI, Penninx BWJH, Wray NR. Disease and polygenic architecture: avoid trio design and appropriately account for unscreened control subjects for common disease. Am J Hum Genet. 2016; 98(2):382–91. https://doi.org/10.1016/j.ajhg.2015.12.017 PMID: 26849113
- Yang J, Zeng J, Goddard ME, Wray NR, Visscher PM. Concepts, estimation and interpretation of SNPbased heritability. Nat Genet. 2017; 49(9):1304. https://doi.org/10.1038/ng.3941 PMID: 28854176
- 42. Sperling RA, Aisen PS, Beckett LA, Bennett DA, Craft S, Fagan AM, et al. Toward defining the preclinical stages of Alzheimer's disease: Recommendations from the National Institute on Aging-Alzheimer's Association workgroups on diagnostic guidelines for Alzheimer's disease. Alzheimer's Dement [Internet]. 2011; 7(3):280–92. Available from: https://doi.org/10.1016/j.jalz.2011.03.003 PMID: 21514248
- Stern Y. What is cognitive reserve? Theory and research application of the reserve concept. 2002. PMID: <u>11939702</u>